Author’s response to reviews

Title: Implementation and evaluation of a Project ECHO telementoring program for the Namibian HIV workforce

Authors:

Leonard Timothy Bikinesi (bikinesi@gmail.com)
Gillian O'Bryan (gilliano@uw.edu)
Clay Roscoe (jclayroscoe@hotmail.com)
Tadesse Mekonen (tadeteferi@gmail.com)
Naemi Shoopala (naemisure@gmail.com)
Assegid Mengistu (mengistua@NACOP.NET)
Souleymane Sawadogo (bya7@cdc.gov)
Simon Agolory (ifz6@cdc.gov)
Gram Mutandi (jiz2@cdc.gov)
Valerie Garises (vgarises@icloud.com)
Rituparna Pati (rpa7@cdc.gov)
Laura Tison (wup7@cdc.gov)
Ledor Igboh (ymq4@cdc.gov)
Carla Johnson (vvlg@aol.com)
Evelyn Rodriguez (eyr6@cdc.gov)
Tedd Ellerbrock (tve1@cdc.gov)
Heather Menzies (fev7@cdc.gov)
Andrew L Baughman (alb1@cdc.gov)
Laura Brandt (laurajbrandt0@gmail.com)
Norbert Forster (norbert@itech-namibia.org)
Thank you for this comment. We thought carefully about the Wilcoxon tests we carried out and your recommendation of a Bonferroni correction. We decided that a more informative and simpler analysis would be to present 95% confidence intervals for the percent improvement in mean knowledge score from the pre-program mean score presented in Table 1. This is consistent with the latest guidance from the American Statistical Association that in most situations, estimation and confidence intervals are more useful than performing null hypothesis statistical testing. Thus,
we have excluded the Wilcoxon tests and added confidence intervals to Table 1, and edited the text accordingly.